Title: Convective burn from use of hairdryer for heel warming prior to the heel prick test - a case report

Authors:

  Robbie Ray (robbie.ray@nhs.net)
  Yvette Godwin (ygodwin@hotmail.com)
  Ashley Shepherd (ashley.shepherd@stir.ac.uk)

Version: 4 Date: 18 April 2011

Author's response to reviews: see over
Dear Editor

Convective burn from use of hairdryer for heel warming prior to the heel prick test – a case report

MS: 9696823495139588

Can I thank reviewer three for his final comments on the above paper. Two minor amendments have been suggested and each has been made to the enclosed paper (in red).

1. In the discussion, paragraph 2, the word ‘oral’ has been added before sucrose
2. In the discussion, paragraph 1, the literature search date limits have been added.

I hope you agree that we have addressed these minor issues. Please do contact me if there are any further points of clarification needed.

Yours sincerely
Dr Ashley Shepherd